Author’s response to reviews

Title: Overexpression of GOLPH3 is associated with poor prognosis and clinical progression in pancreatic ductal adenocarcinoma

Authors:

Luan-Jing Zhang (zhangluanjing@126.com)
Ke-Bing Wang (wkb001788@163.com)
Long-shan Liu (iulongshan2003@163.com)
Lian-zhou Chen (chenzl@126.com)
Bao-Gang Peng (pengbaogang@medmail.com.cn)
Li-Jian Liang (lianjli@medmail.com.cn)
Zhi Li (lizhi@mail.sysu.edu.cn)
Ling Xue (xuel@mail.sysu.edu.cn)
Wen Li (wenli28@163.com)
Jing-Tang Xia (13xia@163.com)

Version: 2
Date: 28 March 2014

Author’s response to reviews: see over
Dear Editor:

I, on behalf of my coauthors, would like you consider the submitted the manuscript entitled “Overexpression of GOLPH3 is associated with poor prognosis and clinical progression in pancreatic ductal adenocarcinoma” for publication in BMC Cancer. Therefore, we examine Golgi phosphoprotein 3 (GOLPH3) expression in 109 cases of formalin-fixed, paraffin-embedded (FFPE) tissue specimens of human pancreatic ductal adenocarcinoma (PDAC), and perform univariate and multivariate analyses to correlate its expression levels with patient survival and clinicopathologic features in PDAC. Our findings suggest that GOLPH3 expression status may be a potential prognostic biomarker and therapeutic target in PDAC.

I can guarantee that the material contained in this manuscript has not been published or submitted to another publisher in part or in whole. I would like to declare that all authors have reviewed the final version of the manuscript and agree to its submission to your journal. Also, there are no conflicts of interest to declare. In advance of submission, we had this manuscript copyedited by a professional English editing service that specializes in scientific papers.

Here, I resubmit a revise manuscript with line and page numbers in the main text.

Finally, I believe that the study fits the scope of your journal and would be interesting to your readers. I hope you find it suitable too, and I hope to hear from you soon.

Sincerely,

Sincerely yours,

Wen Li. M.D., Ph.D.
Laboratory of General Surgery,
The First Affiliated Hospital, Sun Yat-sen University, #58 Zhongshan er Road,
Guangzhou, Guangdong 510080, China
Phone: +86-13728048336  Fax: +86-87755766-8252;
Email: wenli28@163.com; liwen@mail.sysu.edu.cn

Jintang Xia. M.D., Ph.D. MPH